Author’s response to reviews

Title: The Active for Life Year 5 (AFLY5) school based cluster randomised controlled trial: study protocol for a randomized controlled trial.

Authors:

Debbie A Lawlor (d.a.lawlor@bristol.ac.uk)
Russell Jago (russ.jago@bristol.ac.uk)
Sian M Noble (s.m.noble@bristol.ac.uk)
Catherine R Chittleborough (catherine.chittleborough@bristol.ac.uk)
Rona Campbell (rona.campbell@bristol.ac.uk)
Julie Mytton (julie.mytton@bristol.nhs.uk)
Laura Howe (laura.howe@bristol.ac.uk)
Tim Peters (tim.peters@bristol.ac.uk)
Ruth R Kipping (ruth.kipping@bristol.ac.uk)

Version: 3 Date: 11 July 2011

Author’s response to reviews:

We thank the reviewer for their attention to detail
We have corrected the typing error noted by them in the trial flow diagram (figure 1)
In response to the editor we have changed the paper title and ensured it fits with author instructions for the journal.
The award and ethics approval letters have now been removed from this submission.